Author’s response to reviews

Title: Functional analyses of ATM, ATR and Fanconi anemia proteins in lung carcinoma

Authors:

Jan H Beumer (beumerjh@upmc.edu)
Katherine Y Fu (kayfu01@gmail.com)
Bean N Anyang (njohbean@gmail.com)
Jill M Siegfried (jsiegfri@umn.edu)
Christopher J Bakkenist (bakkenistcj@upmc.edu)

Version: 5 Date: 2 July 2015

Author’s response to reviews:

All authors have read and approved the final manuscript and a declaration in the Author Contributions section stating this has been added.

The Headings used and the order of all sections has been checked against the journal author instructions, including the Abbreviations, Competing Interests, Keywords sections etc.